

LSHTM Research Online

James, Philip T; Jawla, Ousubie; Mohammed, Nuredin I; Ceesay, Kabiru; Akemokwe, Fatai M; Sonko, Bakary; Sise, Ebrima A; Prentice, Andrew M; Silver, Matt J; (2019) A novel nutritional supplement to reduce plasma homocysteine in nonpregnant women: A randomised controlled trial in The Gambia. PLoS medicine, 16 (8). e1002870. ISSN 1549-1277 DOI: https://doi.org/10.1371/journal.pmed.1002870

Downloaded from: http://researchonline.lshtm.ac.uk/4654055/

DOI: https://doi.org/10.1371/journal.pmed.1002870

Usage Guidelines:

Please refer to usage guidelines at https://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license: http://creativecommons.org/licenses/by/2.5/

Check for updates

Citation: James PT, Jawla O, Mohammed NI, Ceesay K, Akemokwe FM, Sonko B, et al. (2019) A novel nutritional supplement to reduce plasma homocysteine in nonpregnant women: A randomised controlled trial in The Gambia. PLoS Med 16(8): e1002870. https://doi.org/10.1371/ journal.pmed.1002870

Academic Editor: Huixia Yang, Peking University First Hospital, CHINA

Received: February 27, 2019

Accepted: July 8, 2019

Published: August 13, 2019

Copyright: © 2019 James et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was funded by the UK Medical Research Council (MRC) grant to AMP for the "Impact of maternal diet on the epigenome" (MC_EX_MR/M01424X/1) and a core funding grant to AMP from the MRC/UK Department for the International Development (DFID) Concordat agreement (MC-A760-5QX00). DSM Nutritional RESEARCH ARTICLE

A novel nutritional supplement to reduce plasma homocysteine in nonpregnant women: A randomised controlled trial in The Gambia

Philip T. James[®]*, Ousubie Jawla, Nuredin I. Mohammed[®], Kabiru Ceesay, Fatai M. Akemokwe, Bakary Sonko[®], Ebrima A. Sise, Andrew M. Prentice[®], Matt J. Silver[®]

Medical Research Council Unit The Gambia at the London School of Hygiene and Tropical Medicine, London, United Kingdom

* Philip.James@lshtm.ac.uk

Abstract

Background

Infant DNA methylation profiles are associated with their mother's periconceptional nutritional status. DNA methylation relies on nutritional inputs for one-carbon metabolic pathways, including the efficient recycling of homocysteine. This randomised controlled trial in nonpregnant women in rural Gambia tests the efficacy of a novel nutritional supplement designed to improve one-carbon-related nutrient status by reducing plasma homocysteine, and assesses its potential future use in preconception trials.

Methods and findings

We designed a novel drink powder based on determinants of plasma homocysteine in the target population and tested it in a three-arm, randomised, controlled trial. Nonpregnant women aged between 18 and 45 from the West Kiang region of The Gambia were randomised in a 1:1:1 allocation to 12 weeks daily supplementation of either (a) a novel drink powder (4 g betaine, 800 µg folic acid, 5.2 µg vitamin B12, and 2.8 mg vitamin B2), (b) a widely used multiple micronutrient tablet (United Nations Multiple Micronutrient Preparation [UNIM-MAP]) containing 15 micronutrients, or (c) no intervention. The trial was conducted between March and July 2018. Supplementation was observed daily. Fasted venepuncture samples were collected at baseline, midline (week 5), and endline (week 12) to measure plasma homocysteine. We used linear regression models to determine the difference in homocysteine between pairs of trial arms at midline and endline, adjusted for baseline homocysteine, age, and body mass index (BMI). Blood pressure and pulse were measured as secondary outcomes. Two hundred and ninety-eight eligible women were enrolled and randomised. Compliance was >97.8% for both interventions. At endline (our primary endpoint), the drink powder and UNIMMAP reduced mean plasma homocysteine by 23.6% (-29.5 to -17.1) and 15.5% (-21.2 to -9.4), respectively (both p < 0.001), compared with the controls. Compared with UNIMMAP, the drink powder reduced mean homocysteine by 8.8% (-15.8 to -1.2; p =

Products South Africa provided the UNIMMAP tablets as an in-kind contribution. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AE, adverse event; BMI, body mass index; CI, confidence interval; CV, coefficient of variation; DRI, dietary reference intake; IQR, interquartile range; MRCG, Medical Research Council The Gambia at the London School of Hygiene and Tropical Medicine; MTHFR, methylenetetrahydrofolate reductase; RE, retinol equivalent; SAH, S-adenosyl homocysteine; SAM, S-adenosyl methionine; SD, standard deviation; UNIMMAP, United Nations Multiple Micronutrient Preparation; VA, village assistant; VTRNA2-1, Vault RNA 2–1. 0.025). The effects were stronger at midline. There was no effect of either intervention on blood pressure or pulse compared with the control at endline. Self-reported adverse events (AEs) were similar in both intervention arms. There were two serious AEs reported over the trial duration, both in the drink powder arm, but judged to be unrelated to the intervention. Limitations of the study include the use of a single targeted metabolic outcome, homocysteine.

Conclusions

The trial confirms that dietary supplements can influence metabolic pathways that we have shown in previous studies to predict offspring DNA methylation. Both supplements reduced homocysteine effectively and remain potential candidates for future epigenetic trials in pregnancy in rural Gambia.

Trial registration

Clinicaltrials.gov Reference NCT03431597.

Author summary

Why was this study done?

- At conception, the methylation marks on DNA from sperm and egg are erased and then re-established in the very early embryo.
- This DNA methylation relies on nutritional inputs for one-carbon metabolic pathways, including the efficient recycling of homocysteine, an absence of which might interfere with normal fetal development.
- We sought to test nutritional supplements that could reduce homocysteine and thereby optimise methylation pathways in women at the time of conception.

What did the researchers do and find?

- We designed a novel drink powder containing betaine, folic acid, and vitamins B12 and B2 and tested its efficacy in reducing plasma homocysteine.
- Efficacy was tested in nonpregnant rural Gambian women in a randomised controlled trial, in which we also tested the UNIMMAP supplement that contains 15 micronutrients but no betaine.
- Both our novel supplement and UNIMMAP reduced homocysteine after 5 and 12 weeks of supplementation, with the novel drink powder showing a significantly enhanced efficacy.

What do these findings mean?

- This trial demonstrates that it is feasible to lower homocysteine levels even in women whose homocysteine levels are only marginally elevated.
- Our novel supplement would be a potential candidate for future trials of preconceptional supplements aimed at optimising offspring DNA methylation.

Introduction

Parental nutritional status at the time of conception can influence the lifelong health and disease risk of the developing child [1,2]. Epigenetic modifications provide a plausible mechanism to explain some of these associations [3]. Multiple maternal factors may affect the developing fetal epigenome in early gestation, with nutritional one-carbon-related metabolites being of particular interest [2,4,5]. One-carbon metabolism encompasses metabolic pathways crucial for the provision of methyl groups required for DNA methylation. These pathways involve folate, methionine, serine, glycine, choline, and betaine for methyl group donation, and vitamins B2, B12, and B6 as essential cofactors.

This raises an important public health question: can improving periconceptional nutritional status by providing the nutrients known to support epigenetic processes lead to improved offspring health outcomes? To answer this, a crucial first step is to understand what such a preconception nutritional supplement might look like. This provides the overall rationale for the study we present here.

The intervention we test here targets homocysteine reduction. Homocysteine concentrations amongst women of reproductive age are higher in the Gambian dry season than the rainy season, corresponding with lower concentrations of key one-carbon metabolites such as folate and betaine [6]. In turn, homocysteine is strongly inversely associated with the ratio of S-adenosyl methionine (SAM) to S-adenosyl homocysteine (SAH) [6], a measure of methylation potential [7]. Homocysteine can be hydrolysed to SAH, which acts as an allosteric inhibitor of the SAM to SAH (transmethylation) reaction [8,9]. In our previous research, periconceptional plasma homocysteine concentrations have been strongly inversely associated with infant DNA methylation at several putative metastable epialleles [10-12], genomic loci where methylation is established in the very early embryo [10,13,14]. There is preliminary evidence, for example, that compared with children conceived in the rainy season, those conceived in the dry season demonstrate methylation patterns suggestive of a loss of regular imprinting at the noncoding Vault RNA 2-1 (VTRNA2-1) [12], a locus implicated in immune function and tumour suppression [15,16]. This suggests that infant DNA methylation could be optimised by improving maternal methylation potential in the Gambian dry season through the reduction of homocysteine concentrations.

The effect of a homocysteine-lowering nutritional intervention on infant DNA methylation would need to be tested in a Phase III preconception trial. However, an important first step is to identify the most effective candidate nutritional supplement. We therefore conducted a randomised controlled trial in nonpregnant women, with homocysteine reduction as the primary outcome. We designed a novel drink powder containing betaine and B vitamins tailored specifically to the target population. Recognising that there are also existing, well-accepted multiple micronutrient supplements containing similar ingredients, we tested the drink powder against the United Nations Multiple Micronutrient Preparation (UNIMMAP). Although there are many nutritional intervention studies investigating homocysteine reduction, our study rationale comprised several context-specific questions. First, could a nutritional supplement lower homocysteine in a population where hyperhomocysteinemia is not a known problem? Second, could we specifically reduce Gambian dry season homocysteine concentrations to those found in the rainy season? Third, would a bespoke nutritional supplement have any advantage over a supplement that is already available? By answering these questions, we aimed to identify the most effective candidate nutritional supplement to carry forwards to preconception trials assessing the impact of maternal nutritional status on offspring DNA methylation.

Methods

Trial outline

We conducted a three-arm, parallel, unblinded, randomised, controlled trial with 298 women. Nonpregnant, non-lactating, healthy women of reproductive age from the West Kiang region of The Gambia were randomised in a 1:1:1 allocation ratio to 12 weeks of daily supplementation of either (a) a novel betaine and B vitamins drink powder supplement ('drink powder'), (b) an existing available micronutrient tablet (UNIMMAP), or (c) no intervention (control). All participants were invited for midline (week 5) and endline (week 12) visits for anthropometric measurements and to provide a venepuncture sample. The trial was registered at Clinicaltrials.gov (NCT03431597). The full trial protocol is available in <u>S1 Appendix</u>. The supplementation phase of the trial took place from March to June 2018, in the Gambian dry season.

Our primary objective was to assess the effect of the drink powder on plasma homocysteine after 12 weeks of daily supplementation versus the control group. Our secondary objectives were to look at this comparison at midline, to compare the efficacy of the drink powder versus the UNIMMAP tablet at midline and endline, and to assess any impact of the interventions on blood pressure and pulse versus control at both time points.

Participants

Study participants came from 21 villages of the West Kiang region. A list of potentially eligible women within West Kiang was generated using the Medical Research Council The Gambia at the London School of Hygiene and Tropical Medicine (MRCG) Keneba Health and Demographic Surveillance System database [17]. At the screening stage a team of 10 fieldworkers visited women at their homes to explain the study and eligibility criteria, and to invite them to join the study by providing informed consent.

Women were eligible at the screening stage if they were premenopausal, aged 18–45 years, nonpregnant to their knowledge, non-lactating (at least 9 months postpartum), had no plan to conceive in the ensuing 3 months, had no plans to travel, and with no current illness or chronic health problems (cardiovascular disease, renal disease, thyroid disease, or cancer). Women were excluded if they were currently taking B vitamin or multivitamin supplements or taking medication for the prevention of seizures (e.g., carbamazepine). Women with high blood pressure were excluded only when there was an additional history of stroke or heart attack.

Women of reproductive age in West Kiang region are deficient in several micronutrients central to one-carbon metabolism, and these have been described in detail in previous studies [6,10,18]. For example, almost all women are deficient in B2, over 40% have low vitamin B6 status, approximately 30% have a low concentration of betaine, and 13% are folate deficient [18].

Sample size

Unpublished year-round homocysteine data from a recent trial in the same region [19] indicated that women have a mean (standard deviation [SD]) plasma homocysteine concentration of 8.06 μ mol/L (±2.5) in the peak dry season of February to April. We wanted to detect a minimum change in plasma homocysteine of 1 μ mol/L, which would reduce it to below the mean rainy season concentration (7.31 μ mol/L). This required 99 women per arm to give 80% power (at 5% significance level using a two-sided *t* test), which we increased to 125 to allow for 20% attrition.

Study design

Women who provided informed consent at the screening stage were invited to come for the baseline visit at MRCG Keneba field station. They provided a urine sample for the study nurses to conduct a rapid pregnancy test and, if negative, were formally enrolled onto the trial.

Randomisation. Study participants were randomised to the drink powder, UNIMMAP tablet, or control arm according to a computer-generated randomisation scheme. A list of study IDs was generated in advance by the Keneba database manager. Prior to the baseline visit, the statistician assigned each participant's study ID to one of the three trial arms ('drink powder', 'tablet', 'control') in a 1:1:1 ratio using block randomisation (with a block size of 15). The sub-investigator printed the study ID and group allocation onto a piece of paper and sealed it in an envelope. The envelopes were arranged in consecutive order of participants 1 to 375, and kept for the baseline visit. The sealed envelopes were not an attempt to keep the group allocation blind, but helped ensure that only one person received a study ID and corresponding group allocation at a time upon enrolment, overseen by the field coordinator.

Study visits. At the baseline visit, all enrolled women had their weight (Seca scale model 803, ± 0.1 kg), height (Seca stadiometer model 217 ± 0.1 cm), and blood pressure (Omron M2 automatic monitor) measured. Each measurement was taken in triplicate and the average used. The study nurses collected a fasted venepuncture sample from the participants into 9-mL EDTA monovettes, which were kept on ice. The samples were processed by the laboratory in Keneba within one hour of collection. Samples were spun using an Eppendorf 5810R centrifuge at 1,800 rpm for 10 minutes at 4 °C to separate the plasma. Plasma aliquots were then immediately stored at -70 °C until analysis.

Participants were invited back for a midline visit at week 5 and an endline visit at week 12. The midline visit was programmed at week 5 rather than week 6 to ensure that samples were collected before the start of Ramadan. The exact same procedures were repeated as for base-line. If women were pregnant at midline, they were given the choice of whether to continue in the study or not. Although supplements were safe for pregnancy, the information was recorded to be able to account for any effects of hemodilution at the analysis stage. There were no changes to the originally planned methods after trial commencement.

Design of the novel betaine and B vitamins drink powder. To tailor the supplement to our target population, we analysed three separate data sets containing plasma nutrient concentrations from women of reproductive age in West Kiang. The plasma nutrients were comprised of metabolites having the potential to reduce homocysteine through one-carbon metabolic pathways, many of which having successfully reduced homocysteine in previous intervention studies [20–22]. We ran multivariable linear regression models to assess which nutrients independently predicted plasma homocysteine. We selected those that consistently demonstrated an inverse relationship with homocysteine (folate, B12, B2, betaine) as candidates for the supplement. Full details of these analyses, together with a more detailed discussion of homocysteine metabolism, can be found in S2 Appendix. The chosen supplement

ingredients were then carefully assessed against existing dosage safety information, doses given in previous trials, and current plasma nutrient levels to establish final supplement doses well below upper safety limits. Full details of this process are provided in <u>S3 Appendix</u>.

Supplement composition and supplementation schedule. Table 1 details the composition of the two intervention products. The drink powder contained 4 g of anhydrous betaine with two times the dietary reference intake (DRI) of folic acid, vitamin B12, and vitamin B2 (manufactured by Kendy Ltd., Bankya, Bulgaria, sourcing betaine from Danisco, Naantali, Finland). The UNIMMAP tablet contained 15 micronutrients at the DRI (manufactured by DSM Nutritional Products, Isando, South Africa).

Those participants who were randomised to the intervention arms at baseline started their supplementation the next day in their home villages. The supplements were supplied to participants on a daily basis by village assistants (VAs). VAs were employees of MRCG living in the villages they were supervising. The participants visited their VA on a daily basis. The VAs dissolved the drink powder into 200 mL of water, and participants were given the option to add sugar to taste in case it was too bitter. UNIMMAP was provided in capsule form, to be taken with water. The VAs observed consumption of the supplements. The intensive, daily monitored supplementation schedule meant we termed this trial an efficacy trial.

Once a week, field assistants delivered weekly packs of supplements to the VAs to administer to the participants in their village. The weekly pack included a tally sheet for the VA to record any adverse events (AEs) experienced by the participants daily. They also included seven small coloured cards that were given to the participants, who were asked to give in a card every time they visited the VA in order to measure compliance. Each week the field assistants would compile a report on the AEs from the tally sheets and record compliance, measured by counting the number of cards a participant had returned that week and crosschecking with the number of remaining supplements in the pack. After the 12 weeks of supplementation, the participants continued to be monitored for a further 3 weeks to capture any delayed AEs.

Laboratory methods

We analysed the concentration of plasma homocysteine in samples from baseline, midline, and endline visits at the MRCG Keneba laboratory using the COBAS INTEGRA[®] 400 Plus analyzer. All samples were analysed in one batch on completion of the endline visit. All samples experienced one cycle of freeze-thaw. Inter-assay coefficients of variation (CVs) were <4.0%, and intra-assay CVs were <5.1%.

Blinding

Due to the different characteristics of the drink powder and the tablet, the participants, field assistants, and VAs were not blinded to group allocation. The laboratory technician analysing the plasma homocysteine was blinded to group allocation.

Ethical considerations

The trial received ethics approval from The Gambia Government/MRC Joint Ethics Committee (reference SCC 1575v1.2). All participants provided informed consent by signature, if literate, and otherwise by thumb print in the presence of an impartial witness, if illiterate. The MRCG Clinical Trials Support Office conducted four monitoring trips to ensure the trial was implemented according to the international standards of Good Clinical Practice. All procedures were in accordance with the Helsinki Declaration of 1975 as revised in 1983.

Novel drink powder	UNIMMAP tablet		
Ingredient	Amount in supplement	Ingredient	Amount in supplement
Folate as folic acid	800 μg	Folic acid	400 µg
Vitamin B12 as cyanocobalamin	5.2 μg	Vitamin B12	2.6 µg
Riboflavin (vitamin B2) as Riboflavin-5′-phosphate	2.8 mg	Riboflavin	1.4 mg
Betaine as anhydrous betaine	4 g	-	-
-	-	Vitamin A	800 μg RE
-	-	Vitamin D	5 μg
-	-	Vitamin E	10 mg
-	-	Thiamine	1.4 mg
-	-	Niacin	18 mg
-	-	Vitamin B6	1.9 mg
-	-	Vitamin C	70 mg
-	-	Zinc	15 mg
-	-	Iron	30 mg
-	-	Iodine	150 μg
-	-	Selenium	65 μg
-	-	Copper	2 mg

Table 1. Composition of interventional products, per daily dose.

Abbreviations: RE, retinol equivalent; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.t001

Statistical analysis

Compliance was calculated as a percentage of the total number of cards received for each participant over the trial duration. We summarised AEs by calculating the proportion of participants that had self-reported any AE at any point over the trial duration. We compared the baseline distribution of continuous variables between the three trial arms using the Kruskal-Wallis test. We used the Wilcoxon rank-sum test to compare medians of percentage compliance between the two intervention arms, and the chi-squared/Fisher exact test for proportions of participants reporting morbidities.

We used linear regression models to assess the efficacy of the supplements at lowering homocysteine at midline and endline. We first assessed the homocysteine variables for normality using the Shapiro-Wilk test. Variables were right skewed but satisfied normality assumptions after log transformation. We then fitted linear regression models comparing the mean difference in log homocysteine between pairs of trial arms at midline or endline, adjusted for baseline homocysteine, age, and body mass index (BMI). The β coefficients from the regression models provided an estimate of the log ratio of geometric mean homocysteine concentrations, which we also chose to present as percentage change in the geometric mean using the formula (e^{β} -1)*100. The primary analyses were based on a complete case analysis. We also performed a sensitivity analysis with data restricted to nonpregnant, fasted, compliant (>80% compliance) individuals. All analyses were performed using Stata 15.1 (StataCorp 2017, TX).

Results

The participant flow diagram is shown in Fig 1. Of the 511 women initially approached for the trial, 100 were ineligible, 9 refused to start the study, and 104 did not attend the baseline visit. A total of 298 eligible women were therefore enrolled and randomised to the drink powder, UNIMMAP, and control arms. Although only 10 women officially withdrew over the study





https://doi.org/10.1371/journal.pmed.1002870.g001

time frame (9 before midline and 1 before endline), on the days of the study visits, there were many absentees. At midline there were 16 absentees and 45 at endline, despite those in the supplement arms continuing with the intervention in their villages. There were 298, 273, and 243 plasma samples available for homocysteine analysis at baseline, midline, and endline, respectively. Of these, one (in the UNIMMAP arm at midline) had a homocysteine reading below the detection limit.

Baseline characteristics for the enrolled participants are provided in Table 2. There was no difference in distribution of age, BMI, homocysteine concentration, blood pressure, and pulse of participants between trial arms. There was no difference in any of these baseline parameters between those present and absent at midline and endline follow-up.

The unadjusted homocysteine concentrations at each time point are shown in Table 3 and Fig 2. In the drink powder arm, the geometric mean homocysteine concentration decreased by 2.67 μ mol/L and 2.32 μ mol/L compared with the baseline at midline and endline, respectively. A similar, though attenuated, pattern was observed in the UNIMMAP arm, in which midline and endline mean homocysteine concentrations were 1.28 μ mol/L and 1.22 μ mol/L lower than baseline, respectively.

Table 4 provides the results of the pairwise comparisons between trial arms at endline and midline, adjusted for age, BMI, and baseline homocysteine. Compared with the controls, the drink powder and UNIMMAP reduced mean plasma homocysteine by 23.6% (-29.5 to -17.1) and 15.5% (-21.2 to -9.4), respectively (both p < 0.001). Compared with UNIMMAP, the drink powder reduced the mean homocysteine by 8.8% (-15.8 to -1.2; p = 0.025). Effects at midline were more pronounced, in which the drink powder reduced mean plasma homocysteine by 35.1% (-39.7 to -30.2) compared with the controls, and UNIMMAP reduced homocysteine by 23.1% (-28.1 to -17.8). The drink powder reduced mean homocysteine 14.3% (-20.0 to -8.2) more than UNIMMAP (all *p*-values <0.001).

We performed a sensitivity analysis excluding any participants who became pregnant during the study period, who gave a non-fasted blood sample, or had compliance <80%. At midline, 8 participants were found to be pregnant and decided to continue with the study (2/93 [2.15%] in the control arm, 0 in the drink powder arm, and 6/92 [6.52%] in the UNIMMAP arm). At endline, a total of 19 women were pregnant (4/82 [4.88%] in the control arm, 7/77 [9.09%] in the drink powder arm, and 8/84 [9.53%] in the UNIMMAP arm). At midline, one person reported they had not fasted before venepuncture (in the UNIMMAP arm). At endline, 16 participants had not fasted (1 in control arm, 6 in drink powder arm, 9 in UNIMMAP arm). The trial results excluding pregnant, non-fasted, and noncompliant individuals are provided in <u>S1 Table</u>. Effect sizes and *p*-values are similar to those described in the complete case analysis above.

Blood pressure and pulse results are provided in <u>Table 5</u> and <u>S2 Table</u>. Overall effect sizes were very small, and there was no evidence to suggest that either intervention decreased blood pressure or pulse compared with the controls at endline.

In this trial, supplementation was observed daily, and compliance was extremely high in both intervention arms, at 97.8% for UNIMMAP and 98.8% for the drink powder (Table 6). There was no difference in self-reported AEs between the intervention arms. Table 5 describes the proportion of participants who reported an AE to their VA on at least one day over the 12-week period. The most common AEs described for the two interventions were abdominal pain, nausea, and dizziness, followed by urine discolouration and fever. The frequency of these AEs, however, was low. Mean AE duration across the trial was less than half a day per person, and the maximum number of days an individual reported a given AE was 19 days, for abdominal pain (additional details in S3 Table). There were two serious AEs reported over the trial duration (one miscarriage in week 9 and one stroke in week 10), both in the drink powder

Table 2. Baseline characteristics.

Variable	Control (N = 100); median (IQR)	Drink powder (N = 93); median (IQR)	UNIMMAP (N = 105); median (IQR)	p-Value*
Age (years)	28.5 (22.1–38.7)	30.7 (21.7–39.7)	26 (20.2–38.8)	0.57
BMI (kg/m ²)	21.3 (18.6–25.3)	21.9 (19.6–26.3)	21.0 (19.0–23.5)	0.81
Homocysteine (µmol/L)	9.7 (7.9–12.3)	10.4 (7.6–14.1)	10.2 (7.2–12.7)	0.27
Systolic BP (mmHg)	114 (105–124)	114 (107–122)	114 (105–122)	0.30
Diastolic BP (mmHg)	70 (63–77)	70 (65–77)	69 (62–75)	0.30
Pulse (beats per minute)	72 (65–79)	71 (63–81)	72 (66–77)	0.96

*Kruskal-Wallis test.

Abbreviations: BMI, body mass index; BP, blood pressure; IQR, interquartile range; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.t002

arm. The Trial Safety Monitor and Medical Expert judged both events as highly unlikely to have been caused by the supplement, and both participants resumed participation in the trial after the events.

Discussion

In this trial, both supplements were highly efficacious in reducing homocysteine after 12 weeks of daily supplementation, with the greatest effect sizes apparent after 5 weeks. There was evidence to suggest the drink powder was more efficacious than UNIMMAP at midline and that this advantage was maintained, although diminished, at endline. Trial compliance was high and self-reported AEs were low throughout the trial, and neither varied by intervention arm. The trial confirms an important proof of principle that dietary supplements can influence metabolic pathways that we have shown in previous observational studies to predict offspring methylation levels [10–12].

Nutritional supplements comprising various combinations of folic acid, B vitamins, and methyl donors such as betaine have been successful in lowering plasma homocysteine in multiple trials [20,23–26]. However, despite a call over 10 years ago for more supplementation studies looking at the combined effect of B vitamins with betaine [27], there still remain very few. A meta-analysis of 12 folic acid supplementation trials (with or without additional B vitamins) showed that folic acid supplementation between 1 and 3 months of supplementation reduced homocysteine by 25% (similar across all doses from 0.5 to 5 mg) [20]. The UNIMMAP multivitamin tablet, which contains one DRI of folic acid, B12, B2, and B6, amongst other ingredients, demonstrated similar efficacy in our trial.

The drink powder contained double the amount of folic acid, B12, and B2 compared with UNIMMAP, alongside the addition of betaine, components that may explain its comparatively greater efficacy. Doubling the B vitamin dosage may not greatly contribute towards enhanced homocysteine-reduction properties: previous studies suggest that the dose of folic acid does

Study arm Baseline			Midline		Endline	
	N	Geometric mean (95% CI)	N	Geometric mean (95% CI)	N	Geometric mean (95% CI)
Control	100	10.00 (9.28–10.78)	93	11.50 (10.62–12.46)	82	10.18 (9.54–10.87)
UNIMMAP	105	9.83 (9.10-10.62)	91	8.55 (8.02-9.12)	84	8.61 (8.14–9.12)
Drink powder	93	10.59 (9.67–11.60)	88	7.90 (7.33-8.51)	77	8.27 (7.66-8.92)

Table 3. Unadjusted plasma homocysteine (µmol/L) by time point.

Abbreviations: CI, confidence interval; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.t003



Fig 2. Geometric mean homocysteine over time by trial arm. Error bars = 95% CIs. CI, confidence interval; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.g002

not greatly matter [20], that additional B12 also has a disputed effect [20,28], and that B2 supplementation is only effective amongst those with the rare methylenetetrahydrofolate reductase (*MTHFR*) 677 TT genotype [29,30]. In contrast, in a meta-analysis, 4–6 g of daily betaine supplementation for 6–12 weeks decreased plasma homocysteine by 1.23 µmol/L, corresponding to a 11.8% reduction of mean baseline values [22]. We can therefore speculate that the betaine in the drink powder has an additive effect supplementary to the action of folic acid and B vitamins in UNIMMAP. The results also suggest that the vitamin B6 in UNIMMAP is not required to reduce homocysteine in this population, supporting findings from previous trials,

Table 4.	Multivariable reg	gression results	showing percen	tage difference in	geometric mean	plasma homoc	vsteine between trial arms.
					0		

Trial arms	Time point	N	β (95% CI)*	Percent difference in geometric means**	<i>p</i> -Value
Drink powder versus control [†]	12 weeks	159	-0.27 (-0.35 to -0.19)	-23.6 (-29.5 to -17.1)	< 0.001
UNIMMAP versus control	12 weeks	166	-0.17 (-0.24 to -0.10)	-15.5 (-21.2 to -9.4)	< 0.001
Drink powder versus UNIMMAP	12 weeks	161	-0.09 (-0.17 to -0.01)	-8.8 (-15.8 to -1.2)	0.025
Drink powder versus control	5 weeks	181	-0.43 (-0.51 to -0.36)	-35.1 (-39.7 to -30.2)	< 0.001
UNIMMAP versus control	5 weeks	184	-0.26 (-0.33 to -0.20)	-23.1 (-28.1 to -17.8)	< 0.001
Drink powder versus UNIMMAP	5 weeks	179	-0.15 (-0.22 to -0.09)	-14.3 (-20.0 to -8.2)	< 0.001

* β represents a difference in log homocysteine at a time point between trial arms, adjusted for baseline homocysteine, age, and BMI at the time point. **Calculated from (e^{β} -1)*100.

[†]Primary outcome.

Abbreviations: BMI, body mass index; CI, confidence interval; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.t004

Systolic blood pressure				
Trial arms	Time point	N	β (95% CI)*	<i>p</i> -Value
Drink powder versus control	12 weeks	159	-0.01 (-0.04 to 0.01)	0.233
UNIMMAP versus control	12 weeks	165	0.01 (-0.01 to 0.04)	0.213
Drink powder versus UNIMMAP	12 weeks	160	-0.03 (-0.05 to -0.01)	0.008
Drink powder versus control	5 weeks	181	0.01 (-0.01 to 0.03)	0.511
UNIMMAP versus control	5 weeks	184	0.02 (0.00 to 0.04)	0.024
Drink powder versus UNIMMAP	5 weeks	179	-0.02 (-0.04 to 0.00)	0.104
Diastolic blood pressure			· ·	
Trial arms	Time point	N	β (95% CI)*	p-Value
Drink powder versus control	12 weeks	159	-0.01 (-0.04 to 0.01)	0.329
UNIMMAP versus control	12 weeks	165	0.00 (-0.02 to 0.03)	0.746
Drink powder versus UNIMMAP	12 weeks	160	-0.02 (-0.04 to 0.01)	0.129
Drink powder versus control	5 weeks	181	-0.01 (-0.03 to 0.01)	0.318
UNIMMAP versus control	5 weeks	184	0.00 (-0.02 to 0.02)	0.967
Drink powder versus UNIMMAP	5 weeks	179	-0.01 (-0.03 to 0.01)	0.290
Pulse				
Trial arms	Time point	N	β (95% CI)*	p-Value
Drink powder versus control	12 weeks	159	-0.02 (-0.05 to 0.01)	0.218
UNIMMAP versus control	12 weeks	165	0.00 (-0.04 to 0.03)	0.780
Drink powder versus UNIMMAP	12 weeks	160	-0.02 (-0.05 to 0.02)	0.368
Drink powder versus control	5 weeks	181	0.00 (-0.04 to 0.03)	0.695
UNIMMAP versus control	5 weeks	184	-0.02 (-0.05 to 0.01)	0.222
Drink powder versus UNIMMAP	5 weeks	179	0.01 (-0.02 to 0.04)	0.510

Table 5. Multivariable linear regression results showing difference in blood pressure and pulse between trial arms.

 $^*\beta$ represents the difference in the log-transformed dependent variable between trial arms at each time point, adjusted for the baseline value of the dependent variable, age, and BMI at the time point.

Abbreviations: BMI, body mass index; CI, confidence interval; UNIMMAP, United Nations Multiple Micronutrient Preparation.

https://doi.org/10.1371/journal.pmed.1002870.t005

including those involving participants with low B6 status [20,31,32]. It appears that tailoring the drink powder design to the nutritional profile of the local population using plasma determinants of homocysteine worked well, and could be a viable approach for future supplement design.

The performance of both interventions surpassed our expectations. Given that many of the previous folic acid supplementation trials have been performed in patients with hyperhomocysteinemia, those with increased risk of cardiovascular events, and amongst more elderly participants [20], we had envisaged a more modest reduction of around 1 µmol/L in our healthy, relatively young population. Furthermore, because we could have expected plasma homocysteine to decrease in the controls by endline in mid-June (expected seasonal trend shown in Fig 1 in S1 Appendix), we did not anticipate such a strong effect beyond the seasonal trend. Interestingly, plasma homocysteine concentrations in the controls did not decrease over the trial period. Indeed, the heightened efficacy of the interventions at midline may be driven by higher plasma homocysteine at this time in the controls. A previous longitudinal study of one-carbon biomarkers in nonpregnant women in The Gambia also showed that maximum plasma homocysteine concentration and minimum folate concentration occurred in late March–early April [6], the timing of our midline.

Compliance and morbidity	UNIMMAP	Drink powder	<i>p</i> -Value*
Percent compliance (Median IQR)	97.8 (91.0–100.0)	98.8 (91.6–100.0)	0.23
Nausea (N%)	17/104 (16.4)	8/93 (8.6)	0.10
Dizziness (N%)	14/104 (13.5)	12/93 (12.9)	0.91
Urine discolouration (N%)	4/104 (3.9)	8/93 (8.6)	0.23
Abdominal pain (N%)	19/104 (18.3)	13/93 (14.0)	0.41
Fever (N%)	4/104 (3.9)	6/93 (6.5)	0.52

Table 6. Compliance and self-reported morbidity (condition ever experienced over trial).

*Wilcoxon rank-sum test of medians for percentage compliance, chi-squared test/Fisher exact test for morbidity.

Abbreviations: IQR, interquartile range; UNIMMAP, Daily United Nations Multiple Micronutrient Preparation supplement.

https://doi.org/10.1371/journal.pmed.1002870.t006

Although the drink powder reduced homocysteine more quickly than UNIMMAP at midline, by endline the differences were less pronounced. We have a particular interest in the potential for a micronutrient supplement to influence the establishment of DNA methylation marks in the very early embryo [10,14]. The sensitive periconceptional period may extend back beyond the three months required for the most active phase of oocyte maturation [2], and there is increasing awareness that interventions should cover longer periods than this to account for unknown timing of conception [33]. Therefore, factors such as cost-effectiveness and long-term acceptability become relevant considerations for candidate supplements alongside speed of action. The high compliance rates in both arms suggest that both interventions were acceptable in the village community settings. The low reported morbidity also reinforces the low-risk nature of these supplements. UNIMMAP has been provided in preconception and pregnancy trials in numerous settings [34-42]. Of the trials that reported morbidity, the majority found there to be no difference between UNIMMAP and control arms. Given that the drink powder arm in our trial reported similar levels of AEs to the UNIMMAP arm, we have assurance that both are well tolerated, further supported by the fact that no participant withdrew from the trial on account of AEs.

Lowering homocysteine and providing one-carbon-related nutrients are essential for a range of biological processes related to fertility and embryo development, and epigenetic mechanisms are just one of several aspects to consider at periconception [2,43,44]. Future supplement designs may therefore want to consider adding other nutrients essential for optimal fetal development that were beyond the scope of this trial. Choline, for example, not only acts as a metabolic precursor to betaine but as an essential component of lipids, lipoproteins, and neurotransmitters [45,46]. However, the interrelated nature of metabolic pathways may mean that the provision of supplemental folic acid and betaine is sufficient to reduce the demand for choline for transmethylation and therefore spare it for other important functions [47–49]. Further research is required to determine the extent of such compensatory mechanisms and the implication for supplement design.

Our trial had some limitations. We used homocysteine as an integrated indicator of B vitamin status and methylation potential, but, had resources allowed, we would have liked to quantify the effects of the two supplements on micronutrient status. The study visits had a number of no-shows. However, absentees at midline and endline were not different in baseline characteristics from those who attended. Unfortunately, we were unable to extend the midline visit due to the need to avoid the month of Ramadan, when venepuncture sample collection was not viable. Whilst this meant we did not achieve our intended sample size of 99 women per arm, we were adequately powered to detect the larger reduction in homocysteine observed in our sample. Resource limitations restricted the number of time points we could analyse, but future trials may benefit from considering additional time points and exploring potential nonlinear patterns of homocysteine lowering over time. Conducting this trial earlier in the dry season, for example January to March, when plasma homocysteine concentrations would be expected to be higher than our implementation period of March to June, may have enabled us to see an even stronger impact of the interventions. Practical limitations also meant we did not visit the control group on a daily basis as we did for the intervention groups. Therefore, we did not collect morbidity information from controls, which would have helped us assess the AEs from the interventions relative to the general population. Finally, we did not collect information on genetic polymorphisms that may have affected homocysteine metabolism [50].

In conclusion, both supplements worked to reduce homocysteine concentrations from those seen in the dry season to below those in the rainy season, and therefore met the trial objective. It is premature to speculate whether the improvements on homocysteine reduction offered by the drink powder would lead to significant changes in offspring DNA methylation in future trials. As such, they both remain potential candidates for future epigenetic trials in pregnancy in the rural Gambian setting. This is contingent on future research confirming causal links between maternal nutrient exposure, offspring methylation, and phenotypic effects, which is an area of continued investigation.

Supporting information

S1 Appendix. Clinical trial protocol. (PDF)

S2 Appendix. Determinants of plasma homocysteine as ingredients for the novel drink powder supplement.

(PDF)

S3 Appendix. Safety and dosage considerations for the drink powder supplement. (PDF)

S1 Table. Multivariable regression results showing percentage difference in geometric mean plasma homocysteine between trial arms, restricted to fasted, nonpregnant, compliant participants.

(PDF)

S2 Table. Unadjusted blood pressure and pulse results by time point and trial arm. (PDF)

S3 Table. Additional details on morbidity. (PDF)

S1 Data. Data underlying the findings. (XLS)

S1 CONSORT Checklist. CONSORT checklist. (PDF)

Acknowledgments

The authors thank all the trial participants for their time and commitment and all members of the MRCG field, laboratory, logistic, data, monitoring, and clinical teams for their hard work during implementation. We thank Dr Klaus Kraemer and Anthony Hehir for their guidance on choice of suppliers, and all at Kendy Ltd. for the fast production of the drink powder.

Thank you to Dr Sophie Moore and Prof Patrick Stover for guidance on supplement doses for the drink powder.

Author Contributions

Conceptualization: Philip T. James, Andrew M. Prentice, Matt J. Silver.

Data curation: Philip T. James, Kabiru Ceesay, Bakary Sonko, Ebrima A. Sise.

Formal analysis: Philip T. James.

Funding acquisition: Andrew M. Prentice, Matt J. Silver.

- **Investigation:** Philip T. James, Ousubie Jawla, Kabiru Ceesay, Fatai M. Akemokwe, Andrew M. Prentice.
- Methodology: Philip T. James, Nuredin I. Mohammed, Ebrima A. Sise, Andrew M. Prentice, Matt J. Silver.

Project administration: Philip T. James, Ousubie Jawla, Kabiru Ceesay.

Software: Bakary Sonko.

Supervision: Philip T. James, Ousubie Jawla, Kabiru Ceesay, Fatai M. Akemokwe, Andrew M. Prentice, Matt J. Silver.

Validation: Nuredin I. Mohammed, Bakary Sonko, Ebrima A. Sise, Matt J. Silver.

Writing - original draft: Philip T. James.

Writing – review & editing: Philip T. James, Ousubie Jawla, Nuredin I. Mohammed, Kabiru Ceesay, Fatai M. Akemokwe, Bakary Sonko, Ebrima A. Sise, Andrew M. Prentice, Matt J. Silver.

References

- Fleming TP, Watkins AJ, Velazquez MA, Mathers JC, Prentice AM, Stephenson J, et al. Origins of lifetime health around the time of conception: causes and consequences. Lancet. Elsevier; 2018; 391: 1842–1852. https://doi.org/10.1016/S0140-6736(18)30312-X PMID: 29673874
- Steegers-Theunissen RPM, Twigt J, Pestinger V, Sinclair KD. The periconceptional period, reproduction and long-term health of offspring: the importance of one-carbon metabolism. Hum Reprod Update. 2013; 19: 640–55. https://doi.org/10.1093/humupd/dmt041 PMID: 23959022
- Bianco-Miotto T, Craig JM, Gasser YP, van Dijk SJ, Ozanne SE. Epigenetics and DOHaD: from basics to birth and beyond. J Dev Orig Health Dis. 2017; 8: 513–519. <u>https://doi.org/10.1017/</u> S2040174417000733 PMID: 28889823
- Dominguez-Salas P, Cox SE, Prentice AM, Hennig BJ, Moore SE. Maternal nutritional status, C(1) metabolism and offspring DNA methylation: a review of current evidence in human subjects. Proc Nutr Soc. 2012; 71: 154–65. https://doi.org/10.1017/S0029665111003338 PMID: 22124338
- James P, Sajjadi S, Tomar AS, Saffari A, Fall CHD, Prentice AM, et al. Candidate genes linking maternal nutrient exposure to offspring health via DNA methylation: a review of existing evidence in humans with specific focus on one-carbon metabolism. Int J Epidemiol. 2018; 47: 1910–1937. https://doi.org/10. 1093/ije/dyy153 PMID: 30137462
- Dominguez-Salas P, Moore SE, Cole D, da Costa K-A, Cox SE, Dyer RA, et al. DNA methylation potential: dietary intake and blood concentrations of one-carbon metabolites and cofactors in rural African women. Am J Clin Nutr. 2013; 97: 1217–27. https://doi.org/10.3945/ajcn.112.048462 PMID: 23576045
- Mason JB. Biomarkers of Nutrient Exposure and Status in One-Carbon (Methyl) Metabolism. J Nutr. 2003; 133: 941S–947. https://doi.org/10.1093/jn/133.3.941S PMID: 12612180
- Škovierová H, Vidomanová E, Mahmood S, Sopková J, Drgová A, Červeňová T, et al. The Molecular and Cellular Effect of Homocysteine Metabolism Imbalance on Human Health. Int J Mol Sci. 2016; 17. https://doi.org/10.3390/ijms17101733 PMID: 27775595
- 9. Kerr SJ. Competing methyltransferase systems. J Biol Chem. 1972; 247: 4248–52. PMID: 4338482

- Dominguez-Salas P, Moore SE, Baker MS, Bergen AW, Cox SE, Dyer R, et al. Maternal nutrition at conception modulates DNA methylation of human metastable epialleles. Nat Commun. 2014; 5: 3746. https://doi.org/10.1038/ncomms4746 PMID: 24781383
- Kühnen P, Handke D, Waterland RA, Hennig BJ, Silver M, Fulford AJ, et al. Interindividual Variation in DNA Methylation at a Putative POMC Metastable Epiallele Is Associated with Obesity. Cell Metab. 2016; 24: 502–9. https://doi.org/10.1016/j.cmet.2016.08.001 PMID: 27568547
- Silver MJ, Kessler NJ, Hennig BJ, Dominguez-Salas P, Laritsky E, Baker MS, et al. Independent genomewide screens identify the tumor suppressor VTRNA2-1 as a human epiallele responsive to periconceptional environment. Genome Biol. 2015; 16: 118. https://doi.org/10.1186/s13059-015-0660-y PMID: 26062908
- Van Baak T, Coarfa C, Dugué P, Fiorito G, Laritsky E, Baker M, et al. Epigenetic supersimilarity of monozygotic twin pairs. Genome Biol. 2018; 19:2. https://doi.org/10.1186/s13059-017-1374-0 PMID: 29310692
- Kessler NJ, Waterland RA, Prentice AM, Silver MJ. Establishment of environmentally sensitive DNA methylation states in the very early human embryo. Sci Adv. 2018; 4: eaat2624. <u>https://doi.org/10.1126/ sciadv.aat2624</u> PMID: 30009262
- Lee H-S, Lee K, Jang H-J, Lee GK, Park J-L, Kim S-Y, et al. Epigenetic silencing of the non-coding RNA nc886 provokes oncogenes during human esophageal tumorigenesis. Oncotarget. 2014; 5: 3472–81. https://doi.org/10.18632/oncotarget.1927 PMID: 25004084
- Lee K, Kunkeaw N, Jeon SH, Lee I, Johnson BH, Kang G-Y, et al. Precursor miR-886, a novel noncoding RNA repressed in cancer, associates with PKR and modulates its activity. RNA. 2011; 17: 1076–89. https://doi.org/10.1261/rna.2701111 PMID: 21518807
- Hennig BJ, Unger SA, Dondeh BL, Hassan J, Hawkesworth S, Jarjou L, et al. Cohort Profile: The Kiang-West Longitudinal Population Study (KWLPS)—a platform for integrated research and health care provision in rural Gambia. Int J Epidemiol. 2017; 46:e13. https://doi.org/10.1093/ije/dyv206 PMID: 26559544
- James PT, Dominguez-Salas P, Hennig BJ, Moore SE, Prentice AM, Silver MJ. Maternal One-Carbon Metabolism and Infant DNA Methylation between Contrasting Seasonal Environments: A Case Study from The Gambia. Curr Dev Nutr. 2019; 3: nzy082. <u>https://doi.org/10.1093/cdn/nzy082</u> PMID: 30706047
- Moore SE, Fulford AJ, Darboe MK, Jobarteh ML, Jarjou LM, Prentice AM. A randomized trial to investigate the effects of pre-natal and infant nutritional supplementation on infant immune development in rural Gambia: the ENID trial: Early Nutrition and Immune Development. BMC Pregnancy Childbirth. 2012; 12: 107. https://doi.org/10.1186/1471-2393-12-107 PMID: 23057665
- Homocysteine Lowering Trialists' Collaboration. Lowering blood homocysteine with folic acid based supplements: meta-analysis of randomised trials. BMJ. 1998; 316: 894–8. PMID: 9569395
- Huang T, Zheng J, Chen Y, Yang B, Wahlqvist ML, Li D. High consumption of Ω-3 polyunsaturated fatty acids decrease plasma homocysteine: a meta-analysis of randomized, placebo-controlled trials. Nutrition. 2011; 27: 863–7. https://doi.org/10.1016/j.nut.2010.12.011 PMID: 21501950
- McRae MP. Betaine supplementation decreases plasma homocysteine in healthy adult participants: a meta-analysis. J Chiropr Med. 2013; 12: 20–25. <u>https://doi.org/10.1016/j.jcm.2012.11.001</u> PMID: 23997720
- Steenge GR, Verhoef P, Katan MB. Betaine supplementation lowers plasma homocysteine in healthy men and women. J Nutr. 2003; 133: 1291–5. https://doi.org/10.1093/jn/133.5.1291 PMID: 12730412
- 24. Olthof MR, van Vliet T, Verhoef P, Zock PL, Katan MB. Effect of homocysteine-lowering nutrients on blood lipids: results from four randomised, placebo-controlled studies in healthy humans. PLoS Med. 2005; 2(5): e135. https://doi.org/10.1371/journal.pmed.0020135 PMID: 15916468
- Dawson SL, Bowe SJ, Crowe TC. A combination of omega-3 fatty acids, folic acid and B-group vitamins is superior at lowering homocysteine than omega-3 alone: A meta-analysis. Nutr Res. 2016; 36: 499– 508. https://doi.org/10.1016/j.nutres.2016.03.010 PMID: 27188895
- Olthof MR, Verhoef P. Effects of betaine intake on plasma homocysteine concentrations and consequences for health. Curr Drug Metab. 2005; 6: 15–22. PMID: 15720203
- Benevenga NJ. Consideration of betaine and one-carbon sources of N5-methyltetrahydrofolate for use in homocystinuria and neural tube defects. Am J Clin Nutr. 2007; 85: 946–9. <u>https://doi.org/10.1093/ajcn/85.4.946 PMID: 17413090</u>
- Jacques PF, Bostom AG, Wilson PW, Rich S, Rosenberg IH, Selhub J. Determinants of plasma total homocysteine concentration in the Framingham Offspring cohort. Am J Clin Nutr. 2001; 73: 613–621. https://doi.org/10.1093/ajcn/73.3.613 PMID: 11237940

- 29. McNulty H, Dowey LRC, Strain JJ, Dunne A, Ward M, Molloy AM, et al. Riboflavin Lowers Homocysteine in Individuals Homozygous for the MTHFR 677C→T Polymorphism. Circulation. 2006; 113: 74–80. https://doi.org/10.1161/CIRCULATIONAHA.105.580332 PMID: 16380544
- Hoey L, McNulty H, Strain J. Studies of biomarker responses to intervention with riboflavin: a systematic review. Am J Clin Nutr. Oxford University Press; 2009; 89: 1960S–1980S. https://doi.org/10.3945/ajcn. 2009.27230B PMID: 19403631
- Rios-Avila L, Coats B, Ralat M, Chi Y-Y, Midttun Ø, Ueland PM, et al. Pyridoxine supplementation does not alter in vivo kinetics of one-carbon metabolism but modifies patterns of one-carbon and tryptophan metabolites in vitamin B-6-insufficient oral contraceptive users. Am J Clin Nutr. 2015; 102: 616–25. https://doi.org/10.3945/ajcn.115.113159 PMID: 26201817
- **32.** Lamers Y, Coats B, Ralat M, Quinlivan EP, Stacpoole PW, Gregory JF, et al. Moderate vitamin B-6 restriction does not alter postprandial methionine cycle rates of remethylation, transmethylation, and total transsulfuration but increases the fractional synthesis rate of cystathionine in healthy young men and women. J Nutr. 2011; 141: 835–42. https://doi.org/10.3945/jn.110.134197 PMID: 21430249
- Stephenson J, Heslehurst N, Hall J, Schoenaker DAJM, Hutchinson J, Cade JE, et al. Before the beginning: nutrition and lifestyle in the preconception period and its importance for future health. Lancet. 2018; 391: 1830–1841. https://doi.org/10.1016/S0140-6736(18)30311-8 PMID: 29673873
- Bhutta ZA, Rizvi A, Raza F, Hotwani S, Zaidi S, Moazzam Hossain S, et al. A comparative evaluation of multiple micronutrient and iron-folic acid supplementation during pregnancy in Pakistan: impact on pregnancy outcomes. Food Nutr Bull. 2009; 30: S496–505. <u>https://doi.org/10.1177/15648265090304S404</u> PMID: 20120791
- 35. Liu J, Mei Z, Ye R, Serdula MK, Ren A, Cogswell ME. Micronutrient Supplementation and Pregnancy Outcomes. JAMA Intern Med. 2013; 173: 276.
- Roberfroid D, Huybregts L, Lanou H, Henry M-C, Meda N, Menten J, et al. Effects of maternal multiple micronutrient supplementation on fetal growth: a double-blind randomized controlled trial in rural Burkina Faso. Am J Clin Nutr. 2008; 88: 1330–40. <u>https://doi.org/10.3945/ajcn.2008.26296</u> PMID: 18996870
- Kaestel P, Michaelsen KF, Aaby P, Friis H. Effects of prenatal multimicronutrient supplements on birth weight and perinatal mortality: a randomised, controlled trial in Guinea-Bissau. Eur J Clin Nutr. 2005; 59: 1081–9. https://doi.org/10.1038/sj.ejcn.1602215 PMID: 16015266
- Osrin D, Vaidya A, Shrestha Y, Baniya RB, Manandhar DS, Adhikari RK, et al. Effects of antenatal multiple micronutrient supplementation on birthweight and gestational duration in Nepal: double-blind, randomised controlled trial. Lancet. 2005; 365: 955–62. https://doi.org/10.1016/S0140-6736(05)71084-9 PMID: 15766997
- West KP, Shamim AA, Mehra S, Labrique AB, Ali H, Shaikh S, et al. Effect of maternal multiple micronutrient vs iron-folic acid supplementation on infant mortality and adverse birth outcomes in rural Bangladesh: the JiVitA-3 randomized trial. JAMA. 2014; 312: 2649–58. <u>https://doi.org/10.1001/jama.2014</u>. 16819 PMID: 25536256
- 40. Zagré NM, Desplats G, Adou P, Mamadoultaibou A, Aguayo VM. Prenatal multiple micronutrient supplementation has greater impact on birthweight than supplementation with iron and folic acid: a cluster-randomized, double-blind, controlled programmatic study in rural Niger. Food Nutr Bull. 2007; 28: 317–27. https://doi.org/10.1177/156482650702800308 PMID: 17974365
- Tofail F, Persson LA, El Arifeen S, Hamadani JD, Mehrin F, Ridout D, et al. Effects of prenatal food and micronutrient supplementation on infant development: a randomized trial from the Maternal and Infant Nutrition Interventions, Matlab (MINIMat) study. Am J Clin Nutr. 2008; 87: 704–11. <u>https://doi.org/10. 1093/ajcn/87.3.704</u> PMID: 18326610
- 42. Supplementation with Multiple Micronutrients Intervention Trial (SUMMIT) Study Group, Shankar AH, Jahari AB, Sebayang SK, Aditiawarman, Apriatni M, et al. Effect of maternal multiple micronutrient supplementation on fetal loss and infant death in Indonesia: a double-blind cluster-randomised trial. Lancet. 2008; 371: 215–27. https://doi.org/10.1016/S0140-6736(08)60133-6 PMID: 18207017
- 43. Parisi F, Rousian M, Koning AHJ, Willemsen SP, Cetin I, Steegers EAP, et al. Periconceptional maternal biomarkers of one-carbon metabolism and embryonic growth trajectories: the Rotterdam Periconceptional Cohort (Predict Study). Fertil Steril. 2017; 107: 691–698.e1. https://doi.org/10.1016/j. fertnstert.2016.11.030 PMID: 28069173
- 44. Xu J, Sinclair KD. One-carbon metabolism and epigenetic regulation of embryo development. Reprod Fertil Dev. 2015; 27: 667. https://doi.org/10.1071/RD14377 PMID: 25710200
- Ueland PM. Choline and betaine in health and disease. J Inherit Metab Dis. 2011; 34: 3–15. <u>https://doi.org/10.1007/s10545-010-9088-4</u> PMID: 20446114

- **46.** Jiang X, West AA, Caudill MA. Maternal choline supplementation: a nutritional approach for improving offspring health? Trends Endocrinol Metab. 2014; 25: 263–73. <u>https://doi.org/10.1016/j.tem.2014.02.001 PMID: 24680198</u>
- 47. Hall MN, Howe CG, Liu X, Caudill MA, Malysheva O, Ilievski V, et al. Supplementation with Folic Acid, but Not Creatine, Increases Plasma Betaine, Decreases Plasma Dimethylglycine, and Prevents a Decrease in Plasma Choline in Arsenic-Exposed Bangladeshi Adults. J Nutr. 2016; 146: 1062–7. https://doi.org/10.3945/jn.115.227132 PMID: 27052531
- 48. Obeid R. The metabolic burden of methyl donor deficiency with focus on the betaine homocysteine methyltransferase pathway. Nutrients. 2013; 5: 3481–95. https://doi.org/10.3390/nu5093481 PMID: 24022817
- 49. Bertolo RF, McBreairty LE. The nutritional burden of methylation reactions. Curr Opin Clin Nutr Metab Care. 2013; 16: 102–8. https://doi.org/10.1097/MCO.0b013e32835ad2ee PMID: 23196816
- Stover PJ. Polymorphisms in 1-carbon metabolism, epigenetics and folate-related pathologies. J Nutrigenet Nutrigenomics. 2011; 4: 293–305. https://doi.org/10.1159/000334586 PMID: 22353665